Author’s response to reviews

Title: The current situation of human resources for health in the province of Cabinda in Angola: Is it a limitation to provide universal access to healthcare?

Authors:

Luís Lapão (luis.lapao@ihmt.unl.pt)

Damas Macaia (damasmacaia@hotmail.com)

Version: 2 Date: 17 Apr 2017

Author’s response to reviews:

Title: The current situation of human resources for health in the province of Cabinda in Angola: Is it a limitation to provide universal access to healthcare?

Dear editors

We would like to thank the referees’ pertinent comments and suggestions to the above mentioned paper.

Furthermore, we thank you for taking the time to find a reviewer who provided such a detailed review of our manuscript.

Regarding this review, we have done a complete review the of the text and we have fully addressed all the comments and suggestions – please find all the changes highlighted in the track-changes in the main manuscript and the responses to the major changes in the document attached.